Author's response to reviews

Title: Traditional eye medicine use by newly presenting ophthalmic patients to a teaching hospital in South-eastern Nigeria: socio-demographic and clinical correlates

Authors:

Boniface Ikenna Eze (xy3165767@yahoo.com)
Chimdi Memnufu Chuka-Okosa (chimdi_8691@yahoo.com)
Judith Nkechi Uche (amazingnkachi@yahoo.com)

Version: 6 Date: 16 October 2009

Author's response to reviews: see over
Revisions in this version of the manuscript

1. Authors’ contributions: I have indicated that all the authors read and approved the final manuscript
2. Tables: I have reworked the tables using auto-format template: simple 1.
3. Additional files: I have designated the additional file as recommended
4. References: I have ensured that the numbering of references is in consecutive order.
5. Ethical approval: I have added that the research was in compliance with the Helsinki declaration; ethics: written informed consent was obtained from research participants

    Minor revisions

6. Title: I have used the sentence case in the title, manuscript, and the submission system.
7. Authors’ Institutional affiliations: I have removed this subheading from the title page.
8. Qualifications: I have removed the authors’ qualifications from the manuscript and submission system.
9. Title page: I have removed the authors’ job titles from the title page; I have also created an additional section on About the Authors after authors’ contribution section.
10. I have removed the horizontal line under Table 1.